Author’s response to reviews

Title: Danggwijagyaksan for climacteric syndrome in peri- and postmenopausal women with a blood-deficiency dominant pattern: study protocol for a randomized, double-blind, placebo-controlled pilot trial

Authors:
Mikyung Kim (01mkkim@gmail.com)
Ae-Ran Kim (arkim@kiom.re.kr)
Hyo-Ju Park (mable@kiom.re.kr)
Ojin Kwon (cheda1334@kiom.re.kr)
Joo-Hee Kim (jhkim714@kiom.re.kr)
Eun-Ji Park (pej3153@naver.com)
Seon-Eun Baek (dreamingmong@naver.com)
Jeong-Eun Yoo (jeyoo@dju.ac.kr)
Jun-Hwan Lee (omdjun@kiom.re.kr)

Version: 2 Date: 27 Dec 2017

Author’s response to reviews:

We appreciate your time and effort to review our manuscript. Thanks to your valuable comments and suggestions, we have made some modifications to the manuscript.

The line and page information is based on the tracked copy of the revised manuscript.

Reviewer reports:

Reviewer #2: In the heading Screening: it is better to write in positive way.

For example:
The line: "The subjects who do not meet the following eligibility criteria will not be accepted to this study" can be replaced with the line: "The subjects who meet the following eligibility criteria will be accepted as participants for this study".

=> As per the reviewer’s advice, we revised the screening section in positive way.